Author's response to reviews

Title: Frequency of apnea, bradycardia, and desaturations following first diphtheria-tetanus-pertussis-inactivated polio-Haemophilus influenzae type B immunization in hospitalized preterm infants

Authors:

Jackie Lee (JackieLee@cha.ab.ca)
Joan L Robinson (jr3@ualberta.ca)
Donald W Spady (dspady@ualberta.ca)

Version: 5 Date: 9 May 2006

Author's response to reviews:

Dear Editor,

Thank-you for accepting our manuscript. We have made the following revisions:

1. It is now clarified in the methods that ethics approval was obtained for conducting this study.
2. As suggested by the reviewer, the term "case" was changed to "immunized infant" throughout the manuscript.
3. We have tried to comply with the formatting rules.

Yours sincerely,

Joan L Robinson